Author's response to reviews

Title: Nonattendance in Pediatric Pulmonary Clinics: An Ambulatory Survey

Authors:

Aviv D. Goldbart (avivgold@bgu.ac.il)
Jacob Dreiher (jacod@bgu.ac.il)
Daniel A. Vardy (dvardy@leumit.co.il)
Soliman Alkrinawi (soliman@bgu.ac.il)
Arnon D. Cohen (arcohen@clalit.org.il)

Version: 4 Date: 5 April 2009

Author's response to reviews: see over
April 2 2009

Nicola Collingwood
Editorial Production Assistant
BMC Pulmonary Medicine

Re: Nonattendance in Pediatric Pulmonary Clinics: An Ambulatory Survey
MS: 1666764251210039

Dear Nicola,

Thank you very much for your help. Attached is our revised manuscript. We complied with all formatting changes.

Sincerely,

Dr. Cohen Arnon

Arnon D. Cohen, MD, MPH, PhD
Department of Quality Measures and Research (Director)
Health Planning and Policy Division
General Management
Clalit Health Services
Tel Aviv
ISRAEL

Address for correspondence:
Dr. Arnon D. Cohen
15/2 Haodem St.
Hod Hasharon 45350
ISRAEL
e-mail: arcohen@clalit.org.il
tel: +972-50-6263848
fax: +972-3-7608612